Dear Editor,

Thank you for allowing us to revise our manuscript (Manuscript ID: 94502) entitled "Early diagnosis of esophageal cancer: How to put 'early detection' into effect?". We greatly appreciate the editors' and reviewers' comments and suggestions. These comments and suggestions are valuable for improving our manuscript. All authors approve the revised manuscript for publication.

Response to the reviewer's comments was as follows:

Reviewer #1:

1. Response to comment: This is an article with early diagnosis for esophageal cancer. This article is concise and well organized.
   Response: Thank you for your comment.

Response to the editorial comments was as follows:

Science editor

Response to comment: (1) Country/Territory of origin: China.
(2) The language classification is Grade B. Please visit the following website for the professional English language editing companies that we recommend: https://www.wjgnet.com/bpg/gerinfo/240.
(3) Please add the Running Title: A short running title of no more than 6 words should be provided. Abbreviations are permitted. For example, Losurdo G et al. Two-year follow-up of HCC.
   Response: Thank you for your help. The above content has been confirmed.

Response to comment: (4) The "Key Words" does not meet the requirements: The 'Key words' list will provide 5-10 keywords that reflect the main content of the study. Please do not use abbreviations for the keywords (e.g., Ulcerative colitis, not UC). The first letter of each keyword will be capitalized, and each keyword will be separated by a semicolon, with no terminal period. An example of correct formatting is: Non-alcoholic fatty liver disease; Alcoholic liver disease; Non-alcoholic steatohepatitis; Insulin resistance; Oxidative stress.
   Response: The punctuation between the keywords has been modified as requested.

Response to comment: (5) Reference numbers in the main text. The reference does not require superscripts, please modify it. The correct format is "the spleen hilum[3]."
   Response: The reference numbers have been revised as requested.
Response to comment: (6) References. To ensure the accuracy of the references, please use "Edit References by Auto-Analyser" (https://www.f6publishing.com/Forms/main/ArticleReferenceTool.aspx) to edit the references of the manuscript.

Response: Thanks for your comment. The above content has been checked.

Response to comment: (7) The WJGO article which this editorial discussed has not been listed in the references list. Please add the WJGO article which this editorial discussed into the main text and references list <Qu HT, Li Q, Hao L, Ni YJ, Luan WY, Yang Z, Chen XD, Zhang TT, Miao YD, Zhang F. Esophageal cancer screening, early detection and treatment: Current insights and future directions. World J Gastrointest Oncol 2024; 16: 1180-1191 [DOI: 10.4251/wjgo.v16.i4.1180]>

Response: This article has been added to the main text and references list.

Response to comment: (8) The manuscript must ensure a clean page and no annotations or line numbers are allowed.

Response: Thanks for your comment. The modifications have been made as requested.

Response to comment: (9) Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Response: The above documents have been submitted.

Company editor-in-chief

Response to comment: I have reviewed the Peer-Review Report, full text of the manuscript, all of which have met the basic publishing requirements of the World Journal of Gastrointestinal Oncology, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office's comments and the Criteria for Manuscript Revision by Authors. The author(s) must submit the revised manuscript online through the Intelligent Manuscript Form Editor in F6Publishing system.

Response: Thank you for your comment. The revised manuscript and supplementary documents have been submitted online in the F6Publishing system.
Once again, we would like to express our great appreciation to the editors and reviewers for their comments and suggestions, and we hope the revised manuscript will meet with approval. We sincerely appreciate your consideration of our manuscript.

Best regards,

Jian